Author's response to reviews

Title: DNA repair deficiency in peripheral blood lymphocytes of endometrial cancer patients with a family history of cancer

Authors:

Lyubov Buchynska (lubuchi@mail.ru)
Olga Brieieva (olha.brie@gmail.com)
Nadiia Glushchenko (nadiia-gluko@yandex.ua)
Ludmila Vorobyova (ludmilavorobyova@yandex.ru)
Olena Bilyk (lenabilyk@gmail.com)

Version: 4 Date: 7 October 2014

Author's response to reviews: see over
Dear Editor,

We are pleased to resubmit the revised version of MS: 1843428373114847 Research article “DNA repair deficiency in peripheral blood lymphocytes of endometrial cancer patients with a family history of cancer” to BMC Cancer.

As you notice, we have revised the manuscript by modifying the Discussion sections, based on the comments made by the Referee #3. We also have added gratefulness to Prof. Andrew Collins in Acknowledgments section.

We hope that the revised manuscript is accepted for publication in BMC Cancer.

Kind regards,

Olga Brieieva

Phone: +380442597123

e-mail: olha.brie@gmail.com

Address: RE Kavetsky IEPOR NAS of Ukraine, 45 Vasylkivska st., Kyiv, 03022 Ukraine